Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 23, 2009
Oct. 21, 2009

Injector

· 10/19 The injector drifts sometimes: relative orbit change, transmission goes down, Hall A Compton noise goes up. Do not know the root cause of the drift yet. It can be fixed by steer the orbits back slowly and carefully. 

· After iocin3 was rebooted, neither GUN2 nor GUN3 was selected. If it happens you have to re-select a GUN (we have been using GUN2).
· The injector was optimized. Hall A slit was fully retracted and the Compton noise level was good.
· It appears that the beam in the injector has been drifting but the root cause has not been found yet. The symptoms are high Compton noise in Hall A, high BOM rates in Hall B, and a large energy spread in Hall C. In order to stabilize the beam, Monday the laser spot was moved, injector chopping and de-chopping were optimized, the injector optics was adjusted, and Hall A slit was set to 56mm to reduce the Compton noise. Also two trim cards in the injector were replaced. It appeared that the beam in the injector has been drifting but nothing was identified yet. It is important that we keep relative orbits in the injector. At least we can tell where the beam starts to change.
Accelerator

· 10/21 Site wide power outage around 2100. We are still recovering the machine

· The separators tripped off twice over the weekend on the HV floating deck unit.

· 10/18 Trim Rack SA01B10 power supply was replaced. 

· 10/16 Ray trace data was taken for Hall A to get more optics data to understand the Compton noise.

· (10/16) The Arc1 Box supply failed and was replaced.

· (10/15) Trim Rack BS04B15 Bulk PS was not stable; it was replaced.

· (10/15) The insulating vacuum pump for RF zone 1L03 failed, which caused more helium flow; the faulty pump was replaced.

· (10/15)1L05-3 CM replaced.

· There are many hours of down time due to beam tuning of the injector.
· Thursday (10/8) the CHL tripped due to a failed UPS. 

· MRSEP8A had ELF. The earth sensor did not work properly and was replaced.
· Dogleg 4 did not bend beam enough and the problem was corrected.

· RF separators tripped off a couple of times due to a faulty electronic part.

Hall A---E05-109; pass 3; energy 3.49 GeV
· Hall A current experiment will end on 10/25. A new one will start om 10/30.

· The Compton counts are affected by the injector drifts. It can be corrected by steering orbits back.

· After a Moller run Hall A dipole was left on. The dipole should be off when not doing Moller runs.

· Hall A Moller procedure was updated.
· Hall A Compton noise was high for a few days. The noise level was low after Monday’s tuning. Hall A slit is set to 56mm during the normal running.
Hall B ---Eg6; pass 5; energy 5.764 GeV
· The PD Shift Plan Standing Orders have been updated. Hall B detectors are sensitive to the beam positions. The detectors’ HV would trip off if beam positions are off by a couple of millimeters. Once the HV trips off it would take at least 30 min to get it back. After some accelerator tuning or the hall access, we should ask the hall to turn off its HV prior to sending beam to the hall. When beam current and positions are stable ask the hall to turn the HV back on. It only takes a minute to turn it on or off. 

· Hall B will have a pass change Thursday: from 5 to 1.
· Hall B nA BPM2C21A appears to be working now.
· Hall B beam was good.  

· IPM2C21A may not work properly. EES is investigating it.

· Sometimes Hall B BOM rates are too high, which would trip the BOM FSD. Unstable injector, drifted path length and orbits and RF separation, orbit lock, and the bleedthrough could cause the BOM rates to go high.

· Hall B suggests that Ops not use the slider to adjust its Tagger setting; just type in the setting. Adjusting the slider may trip the Tagger. 

Hall C---HKS; pass 2; energy 2,344 GeV
· Hall C beam is fine.

· Pay attention to Hall C energy spread. If it is more than 3e-5 check the path length and run Krest.

· Hall C is taking low current, 2uA. The ion chamber functionality check was done and the trip levels were reset.

· Hall C energy spread was large for a few days. It was much better after Monday’s injector tuning. Also we can deliver higher current to the hall now (45uA was the highest requested)

· Hall C reported that the frames of their Boron10 and Boron11 targets have burn makes (hall log entry #189760). We do not know how and when it happened yet.

http://opweb.acc.jlab.org/CSUEApps/elog/elog.php
